Author's response to reviews

Title: Intra-articular angioleiomyoma of the knee with an atypical finding of Gd-enhanced magnetic resonance imaging: a case report

Authors:

Taisuke Fukawa (fukawataisuke0625@yahoo.co.jp)
Yorikazu Akatsu (kkaazzuu54@yahoo.co.jp)
Takahisa Sasho (sasho@faculty.chiba-u.jp)

Version: 4 Date: 30 April 2014

Author's response to reviews: see over
Author’s response to reviews

Title: Intra-articular angioleiomyoma of the knee with an atypical finding of Gd-enhanced magnetic resonance imaging: a case report

Authors:

Taisuke Fukawa: fukawataisuke0625@yahoo.co.jp

Yorikazu Akatsu: kkaazzuu54@yahoo.co.jp

Takahisa Sasho*: sasho@faculty.chiba-u.jp

Version: 3 Date: 30 April 2014

Author’s response to reviews: see over
Reviewer's report

Title: Intra-articular angioleiomyoma of the knee with an atypical finding of Gd-enhanced magnetic resonance imaging: a case report

Version: 3 Date: 12 April 2014

Reviewer: Tadashi Tanaka

Comments to authors:
General comments
Thank you for submitting the revised manuscript entitled “Intra-articular angioleiomyoma of the knee with an atypical finding of Gd-enhanced magnetic resonance imaging: a case report”. First of all, I would like to thank you and your co-authors for the considerable efforts you have made with regard to revision of your manuscript. The concerns in the previous manuscript seem to have been adequately addressed, but there are still some deficiencies that were not properly or sufficiently corrected.

In “Abstract”, “Conclusion” has been well changed. However, it does not seem to be satisfactorily done in the body text. In general, it is funny that the content of conclusion of the body text is less (volume) than that of “Abstract”.

We corrected the “Conclusion” as the reviewer indicated.

Quality of written English: Needs some language corrections before being published

Declaration of competing interests:
I declare that I have no competing interests